Author's response to reviews

Title: Sleep complaints and repressive coping in a sample of older Black and White women

Authors:

Girardin Jean-Louis (gjean-louis@downstate.edu)
Carol Magai (cmagai@liu.edu)
Ferdinand Zizi (fzizi@downstate.edu)
Georges J Casimir (gjcrvc@optonline.net)
Jessy Pierre-Louis (pierrelouisj5@yahoo.com)
Pierre M Lubin Jr (pml24@columbia.edu)

Version: 2 Date: 7 November 2006

Author's response to reviews:

Sandra Le, PhD
Senior Assistant Editor
BMC -series journals

Dear Dr. Le,

We appreciate the time devoted to consider our manuscripts for publication by BMC. As suggested by the editor, we agreed to combine the two manuscripts and to have the new manuscript reviewed by BMC Women's Health. The new manuscript is attached.

We look forward to receiving reviewer's comments.

Be well!

Dr. Jean-Louis

Girardin Jean-Louis, Ph.D.
Associate Professor
SUNY Downstate Medical Center (Box 58)
450 Clarkson Avenue
Brooklyn, NY 11203-2098
718-270-4035; 718-604-5734